Author's response to reviews

Title: Outcome of deliveries among adolescent girls at the Yaoundé Central Hospital

Authors:

Florent Ymele Fouelifack (yfouelifack@gmail.com)
Theodore Yangsi Tameh (theotameh@yahoo.co.uk)
Eta Ngole Mbong (mbongeta@yahoo.fr)
Philip Njotang Nana (nananjotangphilip@yahoo.com)
Hortence Jeanne Fouedjio (fouedjiojeanne@yahoo.fr)
Robinson Enow Mbu (rembu2000@yahoo.com)

Version: 4 Date: 8 January 2014

Author's response to reviews:

Fouelifack Ymele Florent. MD
Obstetrician and Gynecologist,
Yaoundé Central Hospital,
PO Box: 31186 Yaoundé – Cameroon;
E-mail: yfouelifack@yahoo.fr
Tel: +237 9631 2354;
29th December 2013.

To the Editor
BMC Pregnancy and Childbirth
Dear Sir/Madam

We hereby submit for your appreciation, the reviewed manuscript of the “Outcome of deliveries among adolescent girls at the Yaoundé Central Hospital” study with the following modifications which took into consideration Dr Lee and Dr Nkwabong’s contributive feedback:

1. We proof-read and improved on the English used.
2. Predictors of perinatal and maternal death were determined.
3. Comparative sub-analyses of pregnancy outcomes in early (10-14), middle (15-17) and late adolescence (18-19 years) were carried out.
4. We added one author whose name is Jovanny Tsuala Fouogue

Thanks for your continuous support, warm regards and wish you a blessed transition into a prosperous and achievements-filled 2014.

Kind regards,
Florent Fouelifack, MD
On behalf of the research team.